Author’s response to reviews

Title: Life-threatening hypersplenism due to idiopathic portal hypertension in early childhood: case report and review of the literature

Authors:

Jan Däbritz (jan.daebritz@ukmuenster.de)
Jennifer Worch (Jennifer.Worch@ukmuenster.de)
Ulrike Materna (umaterna@uni-muenster.de)
Bernward Koch (Bernward.Koch@ukmuenster.de)
Gabriele Koehler (Gabriele.Koehler@ukmuenster.de)
Christina Duck (Christina.Duck@ukmuenster.de)
Michael C Frühwald (michael.fruehwald@klinikum-augsburg.de)
Dirk Foell (dfoell@uni-muenster.de)

Version: 8 Date: 19 October 2010

Author’s response to reviews: see over
MS: 1964654633471521 (Final Version)
Life-threatening hypersplenism due to idiopathic portal hypertension in early childhood: Case report and review of the literature

Dear Mrs Neilan,

Please find enclosed the final version of the above mentioned manuscript for publication in the BMC Gastroenterology. We have addressed the comments in the final manuscript

1. Tables: Tables have been re-layouted without vertical text elements. Shading and colours within the tables were deleted.

2. Typography: The manuscript has been checked for any typographical errors.

Yours sincerely

Dr J. Däbritz
- On behalf of the authors -